\begin{table}%t4
\caption{\label{list}X-ray source list. Count rates are given
per instrument (0.5--7.5~keV in $10^{-1}$~cts/s and in three energy bands 0.5--1.0, 
1.0--2.0, 2.0--7.5~keV).
Sources that are not present in a detector are marked by dashes, sources without useful data by the detector ID in brackets.}
 %\centerline
 { 
\begin{tabular}{lllllllll}\hline\hline
\noalign{\smallskip}
CHXX &RA (h, m, s) & Dec (deg, \arcmin, \arcsec) &Err.(\arcsec) & PN & MOS1 & MOS2 & 2MASS/DENIS-ID$^1$ &Other$^2$ \\\hline
1  & 11~ 07~ 58.1    & --76~ 24~ 43.5  & 3.  &--& --& 0.043 - 0.5/1.8/2.1e-3 & -- & \\
2  & 11~ 08~ 17.6    & --76~ 16~ 35.5  & 3.  &(P) & 0.040 - 2.0/1.8/0.2e-3 & 0.027 - 1.2/1.4/0.1e-3& 2M11081745-7616346 & U0137-0067981\\
3  & 11~ 09~ 02.6    & --76~ 15~ 04.0  & 2.  &0.066 - 3.2/3.1/0.3e-3&0.012 - 5.1/7.2/0.0e-4 & 0.023 - 0.8/1.3/0.2e-3 & 2M11090277-7615034 & U0137-0068083\\
4  & 11~ 09~ 14.2    & --76~ 28~ 41.0  & 2.  &0.16~~ - 8.5/5.9/1.2e-3& -- & 0.048 - 2.1/1.9/0.9e-3& 2M11091380-7628396 & CHXR~35, Hn 8\\
5  & 11~ 09~ 18.0    & --76~ 27~ 59.0  & 2.  &1.21~~ - 5.5/5.2/1.4e-2& -- & 0.37~~ - 1.3/1.8/0.5e-2& 2M11091769-7627578 & CHXR~37, CCE1-74\\
6  & 11~ 09~ 18.4    & --76~ 30~ 30.0  & 2.  &0.22~~ - 0.2/1.2/0.8e-2&-- & 0.11~~ - 0.4/5.3/5.0e-3 & 2M11091812-7630292 & CHXR~79, Hn 9\\
7  & 11~ 09~ 23.7    & --76~ 23~ 22.0  & 2.  &1.07~~ - 3.4/4.5/2.8e-2&-- & 0.36~~ - 0.8/1.7/1.1e-2 & 2M11092379-7623207 & CHXR~39, VZ Cha\\
8  & 11~ 09~ 31.7    & --76~ 18~ 18.5  & 3.  &(P)& -- & 0.012 - 1.5/5.0/5.5e-4& -- & \\
9  & 11~ 09~ 40.3    & --76~ 28~ 40.0  & 1.  &1.52~~ - 7.8/6.0/1.4e-2&0.51~~ - 2.0/2.5/0.6e-1 & 0.46~~ - 1.8/2.2/0.6e-2& 2M11094006-7628391 & CHXR~40, CCE1-78\\
10 & 11~ 09~ 49.3    & --76~ 19~ 32.5  & 3.  &(P)&-- & 0.014 - 0.4/4.7/8.7e-4 & -- &\\
11 & 11~ 09~ 51.3    & --76~ 27~ 16.0  & 2.  &0.025 - 0.6/0.8/1.1e-3&-- & 0.013 - 0.9/6.1/5.7e-4 & -- &\\
12 & 11~ 09~ 54.1    & --76~ 29~ 26.0  & 2.  &0.24~~ - 0.8/1.3/0.4e-2& 0.081 - 1.9/4.3/1.9e-3 & 0.086 - 2.0/4.7/2.0e-3& 2M11095407-7629253 &CHXR~41, Sz 33\\
13 & 11~ 10~ 07.1    & --76~ 29~ 38.5  & 1.  &1.46~~ - 6.4/6.4/1.8e-2& 0.47~~ - 1.6/2.4/0.7e-2 & 0.51~~ - 1.7/2.6/0.8e-2& 2M11100704-7629376 & CHXR~46, WY Cha\\
14 & 11~ 10~ 08.5    & --76~ 15~ 55.0  & 1.  &0.074 - 1.6/3.1/2.7e-3& 0.022 - 5.8/9.3/7.8e-4 & 0.024 - 0.3/1.3/0.9e-3& -- &\\
15 & 11~ 10~ 08.8    & --76~ 19~ 43.0  & 2.  &0.073 - 0.3/2.6/4.4e-3& 0.027 - 0.0/0.9/1.8e-3 & 0.035 - 0.0/1.1/2.4e-3& -- &\\
16 & 11~ 10~ 22.2    & --76~ 25~ 14.0  & 3.  &(P)& -- & 0.012 - 7.1/3.2/1.7e-4& 2M11102226-7625138 & ChaI 710 (BD)\\
17 & 11~ 10~ 22.7    & --76~ 27~ 16.0  & 2.  &0.096 - 1.9/4.1/3.5e-3&-- & 0.041 - 0.5/1.6/2.1e-3 & -- & \\
18 & 11~ 10~ 25.4    & --76~ 20~ 07.0  & 2.  &0.028 - 0.9/1.0/0.9e-3& 0.011 - 2.4/4.8/3.3e-4 & 0.008 - 0.9/4.4/2.8e-4& -- & \\
19 & 11~ 10~ 27.4    & --76~ 27~ 17.5  & 3.  &(P)& -- & 0.011 - 0.1/3.8/7.4e-4& -- &  \\
20 & 11~ 10~ 29.8    & --76~ 23~ 09.0  & 2.  &0.027 - 1.1/1.0/0.5e-3& 0.011 - 3.2/4.4/3.1e-4 & 0.007 - 1.3/3.6/2.1e-4& -- &\\
21 & 11~ 10~ 33.1    & --76~ 20~ 55.0  & 1.  &0.063 - 1.9/2.3/2.1e-3& 0.018 - 6.1/8.2/3.5e-4 & 0.023 - 0.4/1.1/0.8e-3& --& \\
22 & 11~ 10~ 42.2    & --76~ 19~ 13.0  & 1.  &0.14~~ - 4.0/5.8/3.8e-3&0.050 - 1.0/2.4/1.6e-3 & 0.051 - 1.2/2.1/1.8e-3 &-- & \\
23 & 11~ 10~ 50.2    & --76~ 18~ 15.5  & 2.  &(P)&0.029 - 1.5/1.2/0.2e-3 & 0.032 - 1.5/1.3/0.4e-3 & 2M11105039-7618145 &  U0136-0067477\\
24 & 11~ 10~ 52.4    & --76~ 15~ 32.0  & 1.  &0.049 - 0.0/1.4/3.6e-3& 0.011 - 0.3/2.9/7.5e-4 & 0.013 - 0.0/0.2/1.1e-3& --&  \\
25 & 11~ 10~ 52.5    & --76~ 26~ 21.5  & 1.  &0.18~~ - 4.6/7.3/6.0e-3&0.065 - 1.1/3.0/2.4e-3 & 0.061 - 1.3/2.6/2.2e-3 & 2M11105228-7626223 & CHXR~81\\
26 & 11~ 10~ 52.7    & --76~ 28~ 10.5  & 1.  & 0.38~~ - 0.7/1.5/1.7e-2& 0.13~~ - 2.0/5.6/5.1e-3 & 0.11~~ - 1.7/5.1/4.5e-3& -- & CHXR~80 \\
27 & 11~ 10~ 53.2    & --76~ 16~ 56.0  & 2.  &(P)&0.008 - 3.4/4.5/0.3e-4 & 0.009 - 2.4/5.8/1.1e-4 & 2M11105331-7616559 & U0137-0068316 \\
28 & 11~ 10~ 53.7    & --76~ 14~ 29.0  & 2.  &(P)& 0.020 - 3.7/8.9/7.5e-4 & 0.018 - 0.3/1.1/0.5e-3& -- &  U0137-0068321\\
29 & 11~ 10~ 55.6    & --76~ 16~ 29.0  & 1.  & 0.036 - 0.2/1.7/1.6e-3& 0.013 - 0.9/5.4/6.7e-4 & 0.016 - 2.0/6.8/7.3e-4& -- &  \\
30 & 11~ 10~ 57.9    & --76~ 18~ 00.0  & 3.  &(P)&0.011 - 3.1/5.2/2.9e-4 & 0.012 - 3.1/5.5/2.9e-4& 2M11105830-7617580  & U0137-0068332 \\
31 & 11~ 11~ 04.2    & --76~ 24~ 28.5  & 3.  &(P)&0.008 - 0.0/2.0/5.8e-4 & 0.008 - 0.0/2.2/5.7e-4 & -- & \\
32 & 11~ 11~ 13.8    & --76~ 20~ 02.0  & 2.  &0.034 - 1.0/1.4/1.1e-3& 0.012 - 1.2/7.2/3.8e-4 & 0.017 - 3.8/6.9/6.1e-4& -- & \\
33 & 11~ 11~ 30.7    & --76~ 20~ 03.5  & 2.  &0.068 - 4.1/2.6/0.3e-3& 0.022 - 1.0/1.0/0.1e-3 & 0.020 - 0.9/1.1/0.0e-3& 2M11113049-7620030 &  U0136-0067600\\
34 & 11~ 11~ 39.6    & --76~ 23~ 07.0  & 2.  &0.020 - 4.7/6.1/8.8e-4& 0.011 - 1.4/6.8/2.3e-4 & 0.007 - 1.9/4.1/1.5e-4& --& \\
35 & 11~ 11~ 39.7    & --76~ 20~ 15.5  & 1.  &0.59~~ - 2.3/2.7/0.9e-2& 0.19~~ - 0.6/1.0/0.3e-2 & 0.19~~ - 0.6/1.0/0.3e-2& 2M11113965-7620152 & XX Cha\\
36 & 11~ 11~ 44.2    & --76~ 12~ 49.0  & 2.  &(P)& 0.018 - 5.1/5.0/8.0e-4 & 0.013 - 2.3/4.7/5.5e-4& --&  \\
37 & 11~ 11~ 46.4    & --76~ 20~ 09.0  & 1.  & 2.51~~ - 1.1/1.1/0.4e-1& 0.84~~ - 2.6/4.3/1.4e-2 & 0.79~~ - 2.4/4.2/1.3e-2& 2M11114632-7620092& CHX~18N, CCE1-103\\
38 & 11~ 11~ 49.1    & --76~ 30~ 31.0  & 2.  & 0.074 - 1.8/3.1/2.5e-3& 0.020 - 0.2/1.2/0.7e-3 & 0.021 - 0.2/1.2/0.7e-3& -- & \\
39 & 11~ 11~ 54.0    & --76~ 19~ 31.0  & 1.  & 4.29~~ - 2.2/1.6/0.5e-1& 1.33~~ - 5.5/5.9/1.9e-2 & 1.36~~ - 5.2/6.5/2.0e-2& 2M11115400-7619311& Hn 15 \\
40 & 11~ 12~ 03.9    & --76~ 26~ 10.0  & 2.  &0.021 - 4.5/9.0/7.5e-4& 0.005 - 0.8/4.1/4.8e-4 & 0.012 - 0.8/3.4/7.5e-4 &--&  \\
41 & 11~ 12~ 05.3    & --76~ 19~ 00.5  & 2.  &(P)&0.013 - 0.2/5.8/7.1e-4 & 0.013 - 0.0/3.4/9.4e-4 & --& \\
42 & 11~ 12~ 05.6    & --76~ 31~ 38.0  & 2.  &0.13~~ - 2.3/5.8/4.5e-3& 0.046 - 0.2/2.1/2.3e-3 & 0.051 - 0.6/2.7/1.8e-3& --& \\
43 & 11~ 12~ 06.4    & --76~ 19~ 31.0  & 2.  &(P)& 0.019 - 4.9/7.4/6.7e-4 & 0.015 - 3.3/8.6/3.2e-4& --& U0136-0067713\\
44 & 11~ 12~ 09.6    & --76~ 34~ 37.0  & 3.  &0.061 - 1.8/3.3/0.9e-3&(M1) & -- & 2M11120984-7634366 & CHXR~85, Sz 40\\
45 & 11~ 12~ 17.5    & --76~ 19~ 16.0  & 2.  &(P)& 0.014 - 3.3/6.4/3.9e-4 & 0.012 - 1.4/7.8/3.2e-4& --& \\
46 & 11~ 12~ 19.1    & --76~ 17~ 40.5  & 2.  &0.020 - 1.1/0.9/0.0e-3& 0.006 - 2.1/2.4/1.1e-4 & 0.008 - 2.0/5.2/0.6e-4& 2M11121884-7617400 &  U0137-0068518\\
47 & 11~ 12~ 20.1    & --76~ 31~ 44.0  & 3.  &0.048 - 1.0/1.6/2.2e-3& 0.015 - 2.6/5.7/6.2e-4 & 0.017 - 2.5/7.3/6.7e-4& -- &  \\
48 & 11~ 12~ 22.2    & --76~ 18~ 38.5  & 1.  &0.16~~ - 4.0/6.8/5.3e-3&0.047 - 1.0/2.0/1.7e-3 & 0.046 - 1.1/1.9/1.6e-3 & --& \\
49 & 11~ 12~ 22.9    & --76~ 23~ 08.0  & 1.  &0.043 - 0.8/2.1/1.3e-3&0.013 - 1.4/6.6/5.1e-4 & 0.014 - 1.0/5.9/7.0e-4 & --&  \\
50 & 11~ 12~ 27.8    & --76~ 25~ 29.5  & 1.  &0.83~~ - 5.9/2.1/0.3e-2& 0.24~~ - 1.5/0.8/0.1e-2 & 0.26~~ - 1.6/0.9/0.1e-2& 2M11122775-7625293& CHXR~53, CCE1-117\\
51 & 11~ 12~ 34.8    & --76~ 26~ 54.0  & 2.  &0.058 - 1.3/1.8/2.7e-3& 0.018 - 3.1/9.3/5.2e-4 & 0.016 - 0.2/1.0/0.4e-3& -- & \\
52 & 11~ 12~ 40.3    & --76~ 08~ 20.5  & 2.  &0.024 - 1.3/0.7/0.4e-3& 0.014 - 5.2/5.7/3.5e-4 & 0.011 - 3.5/5.9/1.7e-4& 2M11124044-7608195 & U0138-0069520 \\
53 & 11~ 12~ 41.2    & --76~ 10~ 55.5  & 2.  &0.032 - 0.2/1.5/1.6e-3& 0.008 - 0.0/5.7/2.2e-4 & 0.013 - 1.1/4.6/7.1e-4&--& \\
54 & 11~ 12~ 53.0    & --76~ 13~ 55.0  & 2.  &0.026 - 0.3/1.0/1.3e-3& 0.008 - 0.4/1.4/6.6e-4 & -- & --& \\
55 & 11~ 13~ 24.6    & --76~ 29~ 22.5  & 2.  &0.24~~ - 1.4/0.8/0.1e-2& 0.064 - 3.1/3.0/0.4e-3 & 0.062 - 3.2/2.4/0.6e-3& 2M11132446-7629227& CHXR~60SW, Hn 18\\
56 & 11~ 13~ 27.1    & --76~ 31~ 46.0  & 2.  &0.63~~ - 1.4/2.7/2.2e-2& 0.25~~ - 0.4/1.1/0.9e-2 & 0.21~~ - 3.3/9.9/8.2e-3 & D111327.02-763147.0& CHXR 58\\
57 & 11~ 13~ 27.5    & --76~ 20~ 17.0  & 2.  &(P)& 0.024 - 1.4/0.8/0.1e-3 & 0.023 - 1.3/0.8/0.2e-3& 2M11132747-7620175 & U0136-0067981\\
58 & 11~ 13~ 27.8    & --76~ 34~ 17.0  & 2.  &11.4~~ - 4.7/4.5/2.1e-1& -- &  --& 2M11132737-7634165 & CHXR~59, CCE1-117\\
59 & 11~ 13~ 28.1    & --76~ 13~ 10.5  & 2.  &0.069 - 0.1/2.5/4.2e-3& 0.022 - 0.1/0.8/1.4e-3 & 0.025 - 0.1/0.8/1.6e-3& --& \\
60 & 11~ 13~ 29.6    & --76~ 29~ 01.0  & 1.  &0.38~~ - 2.1/1.5/0.2e-2&0.11~~ - 5.3/4.9/1.0e-3 & 0.11~~ - 5.9/4.1/0.9e-3 & 2M11132970-7629012&CHXR~60NE,  Hn 19\\
61 & 11~ 13~ 34.2    & --76~ 15~ 38.0  & 2.  &(P)&0.010 - 0.8/4.8/4.8e-4 & 0.011 - 2.3/5.5/3.5e-4 & --&  \\
62 & 11~ 13~ 46.2    & --76~ 26~ 46.0  & 2.  &0.030 - 0.5/1.0/1.5e-3& -- & 0.013 - 0.4/6.2/6.3e-4& --&  \\
63 & 11~ 14~ 05.6    & --76~ 18~ 51.5  & 2.  &0.052 - 1.4/1.7/2.1e-3& 0.019 - 2.8/6.3/9.4e-4 & 0.016 - 3.0/5.1/8.0e-4& --& \\
64 & 11~ 14~ 15.2    & --76~ 21~ 41.5  & 2.  &0.029 - 0.6/1.4/0.9e-3& 0.010 - 1.0/6.6/2.6e-4 & 0.007 - 1.3/4.9/1.0e-4& --& \\
65 & 11~ 14~ 15.5    & --76~ 27~ 37.0  & 2.  &0.18~~ - 1.1/0.6/0.1e-2& 0.070 - 3.0/3.1/0.8e-3 & 0.055 - 2.1/2.8/0.5e-3& 2M11141565-7627364& CHXR~62, Hn 20\\
66 & 11~ 14~ 29.0    & --76~ 11~ 48.0  & 3.  &0.060 - 1.1/2.8/2.1e-3& 0.020 - 0.3/1.0/0.7e-3 & 0.014 - 0.9/8.9/4.5e-4 &--& U0138-0069824 \\
67 & 11~ 14~ 29.1    & --76~ 25~ 40.0  & 2.  &0.041 - 3.1/1.0/0.0e-3& 0.014 - 8.0/3.1/2.5e-4 & 0.009 - 6.2/2.2/0.9e-4& 2M11142906-7625399 &U0135-0067114\\
58 & 11~ 14~ 42.5    & --76~ 10~ 19.0  & 3.  &0.056 - 1.5/2.4/1.8e-3& -- & --& --& \\
69 & 11~ 14~ 46.5    & --76~ 24~ 14.0  & 2.  &0.051 - 1.1/1.8/2.3e-3& 0.021 - 0.2/1.1/0.8e-3 & 0.018 - 0.0/0.8/1.1e-3& --& \\
71 & 11~ 14~ 49.3    & --76~ 22~ 49.0  & 2.  &0.078 - 1.9/3.7/2.2e-3& 0.024 - 0.5/1.1/0.9e-3 & 0.029 - 0.2/1.3/1.5e-3& --& \\
71 & 11~ 15~ 11.0    & --76~ 20~ 24.0  & 3.  &0.033 - 0.8/1.5/1.0e-3& -- & 0.020 - 2.8/8.1/8.8e-4& --&\\\hline
\end{tabular}}
\par
\smallskip
 1: 2M for 2MASS~J, D for DENIS database 3rd release; 
2: CHXR from \cite{fei93}, Hn from \cite{har93}, Sz from \cite{sch77}, CCE from \cite{cam98}, ChaI from \cite{lop04} and U for USNO-B1.0 catalog.
\end{table}